Author’s response to reviews

Title: Using Hashimoto thyroiditis as gold standard to determine the upper limit value of thyroid stimulating hormone in a Chinese cohort

Authors:

Yu Li (tr13501082964@163.com)
Dong-Ning Chen (chendn@trhos.com)
Jin Cui (trcuijing@163.com)
Guang-Ran Yang (yanggr_55@126.com)
Zhong Xin (xinz@medmail.com.cn)
Ming-Jia Niu (fyynmj@163.com)
Jin-Kui Yang (jkyang@ccmu.edu.cn)

Version: 3 Date: 27 Sep 2016

Author’s response to reviews:

Dear Editor,

Thanks for the reviewer. According to the reviewer’s comment, we added all the compulsory information under the heading “declaration”. Please see the detailed content within the manuscript.

Jin-Kui